Author's response to reviews

Title: Thyroid abscess due to Aspergillosis: a case report

Authors:

Suemi Marui (suemimar@usp.br)
Ariella C.L. Pereira (ariellalima@yahoo.com.br)
Raquel M.A. Maia (uelxy@hotmamail.com)
Eduardo F Borba (eborba@lim17.ffm.usp.br)

Version: 3 Date: 2 April 2014

Author's response to reviews: see over
Thank you for your email and the opportunity to submit a revised version of our manuscript.

We greatly appreciated the constructive comments and suggestions and have fully revised the manuscript as suggested by the reviewers.

We highlighted all changes made and submitted the paper to English review.

Best regards,

Suemi Marui
Author's response to review
Title: Thyroid abscess due to Aspergillosis: a case report
Version: 2
Date: 21 February 2014
Authors:
Suemi Marui: suemimar@usp.br
Ariella Cabral de Lima Pereira: ariellalima@yahoo.com.br
Raquel Maria de Araújo Maia: quelxy@hotmail.com
Eduardo Ferreira Borba: eborba@lim17.fm.usp.br

Thank you for your constructive comments, which have improved our manuscript.

We included your suggestions and reference in this new version.

The role of surgery was included in the Discussion and highlighted.
Author’s response to review
Title: Thyroid abscess due to Aspergillosis: a case report
Version: 2 Date: 21 February 2014
Authors:
Suemi Marui: suemimar@usp.br
Ariella Cabral de Lima Pereira: ariellalima@yahoo.com.br
Raquel Maria de Araújo Maia: quelxy@hotmail.com
Eduardo Ferreira Borba: eborba@lim17.fm.usp.br

We thank you for your comments and important observations. We have submitted this paper for English assistance and orthographic revision.

1. We added and highlighted the cytological characteristic of FNA sample and the *Aspergillus* spp.
2. Amphotericin B and fluconazole is known to have fewer efficacies on Aspergillus treatment than voriconazole, but this association can be done [1].
3. Unfortunately detailed histopathological examination was not provided.
4. Ethical permission was obtained as assured in Author’s page.

References